Author's response to reviews

Title: Tongue lesions in psoriasis: A controlled study

Authors:

Maryam Daneshpazhooh (maryamdanesh@yahoo.com)
Homayoon Moslehi (moslehi_h@yahoo.com)
Maryam Akhyani (ma_akhyani@yahoo.com)
Marjan Etesami (marjanetesami@yahoo.com)

Version: 3 Date: 28 October 2004

Author’s response to reviews:

October 28, 2004

Dear Editors,

Regarding the requested formatting changes, I have made the changes as follows:

- Title: TONGUE LESIONS IN PSORIASIS: A CONTROLLED STUDY....changed to ....Tongue lesions in psoriasis: A controlled study
- Abstract: All colons removed from the subheadings
- Affiliations: repeat affiliations are removed, superscript numbers are removed, the email addresses and the corresponding author information moved to below the affiliation
- Section subheadings: colons removed
- Manuscript text: references placed before the full stops

Competing interests: "The authors declare that they have no competing interests"

I wish the manuscript will be judged ready for publication.

Looking forward to hearing from you

Sincerely yours,

Daneshpazhooh